## PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (http://bmjopen.bmj.com/site/about/resources/checklist.pdf) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

## **ARTICLE DETAILS**

TITLE (PROVISIONAL)	The Canada-DONATE Study Protocol: A Prospective National
	Observational Study of the Medical Management of Deceased
	Organ Donors
AUTHORS	D'Aragon, Frédérick; Dhanani, Sonny; Lamontagne, Francois; Cook,
	Deborah; Burns, Karen; Akhtar, Aemal; Chasse, Michaël; Frenette,
	Anne-Julie; Keenan, Sean; Lize, Jean Francois; Kutsogiannis,
	Demetrios J; Kramer, Andreas H.; Hand, Lori; Arseneau, Erika;
	Masse, Marie-Helene; Ribic, Christine; Ball, Ian; Baker, Andrew;
	Boyd, Gordon; Rochwerg, Bram; Healey, Andrew; Hanna, Steven;
	Guyatt, Gordon; Meade, Maureen

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.